

# **Geophysical Research Letters**<sup>•</sup>

## **RESEARCH LETTER**

10.1029/2024GL111250

#### **Key Points:**

- Up to a four-fold increase in the magnitude of Antarctic freshwater discharge is projected by 2300 for an extreme climate scenario
- We project a shift in the form and depth of Antarctic freshwater export, as subshelf (and potentially surface) melt outpaces solid calving
- Our key findings align with satellitebased observations and are robust despite uncertainties in climate and icedynamical response

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

V. Coulon, violaine.coulon@ulb.be

#### Citation:

Coulon, V., De Rydt, J., Gregov, T., Qin, Q., & Pattyn, F. (2024). Future freshwater fluxes from the Antarctic ice sheet. *Geophysical Research Letters*, *51*, e2024GL111250. https://doi.org/10.1029/ 2024GL111250

Received 12 JUL 2024 Accepted 21 NOV 2024

#### **Author Contributions:**

Conceptualization: Violaine Coulon, Jan De Rydt, Frank Pattyn Formal analysis: Violaine Coulon Funding acquisition: Jan De Rydt, Frank Pattyn Methodology: Violaine Coulon, Jan De Rydt, Thomas Gregov, Frank Pattyn Supervision: Jan De Rydt, Frank Pattyn Validation: Violaine Coulon Visualization: Violaine Coulon Writing – original draft: Violaine Coulon Writing – review & editing: Jan De Rydt, Thomas Gregov, Qing Qin, Frank Pattyn

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# **Future Freshwater Fluxes From the Antarctic Ice Sheet**

Violaine Coulon<sup>1</sup>, Jan De Rydt<sup>2</sup>, Thomas Gregov<sup>1,3</sup>, Qing Qin<sup>2</sup>, and Frank Pattyn<sup>1</sup>

<sup>1</sup>Laboratoire de Glaciologie, Université libre de Bruxelles (ULB), Brussels, Belgium, <sup>2</sup>Department of Geography and Environmental Sciences, Northumbria University, Newcastle, UK, <sup>3</sup>Aérospatiale et Mécanique, Université de Liège (ULiège), Liège, Belgium

**Abstract** Surface freshening of the Southern Ocean driven by meltwater discharge from the Antarctic ice sheet has been shown to influence global climate dynamics. However, most climate models fail to account for spatially and temporally varying freshwater inputs from ice sheets, introducing significant uncertainty into climate projections. We present the first historically calibrated projections of Antarctic freshwater fluxes (subshelf melting, calving, and surface meltwater runoff) to 2300 that can be used to force climate models lacking interactive ice sheets. Our findings indicate substantial changes in the magnitude and partitioning of Antarctic freshwater discharge over the coming decades and centuries, particularly under very-high warming scenarios, driven by the progressive collapse of the West Antarctic ice shelves. We project a shift in the form and location of Antarctic freshwater sources, as liquid sub-shelf melting increases under the two climate scenarios considered, and surface meltwater runoff could potentially become a dominant contributor under extreme atmospheric warming.

**Plain Language Summary** Melting Antarctic ice releases freshwater into the Southern Ocean, which can have profound impacts on the regional and global climate. In a warming world, the freshwater input into the ocean is expected to increase. However, climate models often poorly represent ice-sheet mass loss, leading to uncertainty in climate predictions. This study provides new projections of Antarctic freshwater discharge, including contributions from melting at the base of the floating ice shelves, iceberg calving and surface meltwater runoff, up to the year 2300. These projections can be integrated into climate models that lack interactive ice sheets. The results indicate substantial changes in the amount and nature of freshwater discharge from Antarctica in the coming decades and centuries, especially under extreme warming scenarios. We show that liquid melting beneath the floating ice shelves will increase under the two climate scenarios considered, and that surface meltwater runoff could become a major source of freshwater under very high atmospheric warming conditions.

### 1. Introduction

The Antarctic ice sheet gains mass from snow accumulation in its interior ( $\sim 2500 \text{ Gt yr}^{-1}$ ; Mottram et al., 2021) and loses mass mainly through ice discharge across the grounding line into the ocean, where ice shelves form. These ice shelves, the floating extensions of the ice sheet in direct contact with the ocean, progressively melt or calve icebergs, releasing freshwater into the Southern Ocean. Currently, this freshwater discharge primarily occurs through ocean-induced basal melting underneath the ice shelves (liquid discharge) and iceberg calving at the ice-shelf front (solid discharge). More specifically, approximately half of the ice-sheet surface mass gain is lost through oceanic melting before reaching the calving front (Adusumilli et al., 2020; Davison et al., 2023; Depoorter et al., 2013; Greene et al., 2022; Paolo et al., 2023). Basal melting under ice shelves is an important source of freshwater at depth, influencing Antarctic Bottom Water formation (Silvano et al., 2018), a key component of the global ocean circulation. Icebergs, on the other hand, transport freshwater far beyond the continent, slowly releasing it into the ambient ocean. This contributes to both thermodynamic and dynamic effects on the ocean, which feedback into the atmosphere and influence climate variability (Schloesser et al., 2019). Although all Antarctic ice shelves experience surface melting (Trusel et al., 2013, 2015), only a small amount of mass is currently lost by surface meltwater runoff (Bell et al., 2017). Further minor freshwater contributions from Antarctica are subglacial melt by geothermal and deformational heat underneath the grounded ice sheet (estimated at 65 Gt yr<sup>-1</sup>; Pattyn, 2010), as well as blowing snow that is transported by the wind from the ice-sheet margin onto the ocean (Lawrence et al., 2023).

Surface freshening of the Southern Ocean has been observed in recent decades (de Lavergne et al., 2014). This could in part be due to the Antarctic ice sheet and ice shelves losing mass at an accelerating rate, particularly in



West Antarctica (Adusumilli et al., 2020; Davison et al., 2023; Otosaka et al., 2023; Smith et al., 2020), releasing increasing amounts of meltwater into the ocean. Antarctic mass loss is projected to further accelerate in the future (Edwards et al., 2021; Payne et al., 2021; Seroussi et al., 2020, 2024), making the resulting freshwater input a major contributor to global mean sea-level rise in the coming decades and centuries (Fox-Kemper et al., 2021). Especially, Antarctic ice-shelf basal melt, iceberg calving, and surface runoff are all expected to increase in a warming climate (Coulon, Klose, et al., 2024; Golledge et al., 2019; Trusel et al., 2015).

Such meltwater and ice discharge from the Antarctic ice sheet have been shown to impact global climate dynamics and influence the trajectory of future climate change (Bronselaer et al., 2018; Golledge et al., 2019; Purich & England, 2023; Sadai et al., 2020). This underscores the need to integrate these feedbacks into climate model projections. However, fully coupled climate-ice sheet models have only recently been developed (Pelletier et al., 2022; Siahaan et al., 2022), and most existing climate models do not have interactive ice sheets. Even for CMIP6 models, it is either assumed that ice sheets are in mass balance, or that discharge from the ice sheets remains constant in time (Swart et al., 2023). As a result, most climate projections lack a representation of spatiotemporal trends in ice-sheet freshwater fluxes and, consequently, the impact of (changes in) meltwater and ice discharge from Antarctic ice sheets and ice shelves on the global climate system is not reliably represented. Those who have addressed the critical question of understanding the future influence of Antarctic meltwater on regional and global climate have often used idealized freshwater perturbation experiments (Swart et al., 2023), for example, by approximating the future spatiotemporal evolution of the Antarctic freshwater sources in a warming climate by scaling of observed ratios (e.g., Gorte et al., 2023; Swart et al., 2023; Thomas et al., 2023). This introduces unquantified biases and uncertainties into future climate and sea-level projections (Bronselaer et al., 2018; Golledge et al., 2019; Purich & England, 2023; Sadai et al., 2020). Furthermore, given the significant uncertainty in projected Antarctic mass loss (Fox-Kemper et al., 2021), it is crucial to explore the response and sensitivity of the climate system to a range of realistic (in space and time) changes in meltwater amounts in a warming climate (Purich & England, 2023).

To address this, we present historically calibrated projections of the spatial variability of sub-shelf melt, iceberg calving, and surface runoff from the Antarctic ice sheet under two climate scenarios (a low and a very-high emission scenario) from present-day to 2300. These distributions of future Antarctic freshwater fluxes were produced by considering a wide range of climate and ice-dynamical uncertainties, including ice-shelf melting, iceberg calving, basal sliding and ice damage. The resulting data set hence provides estimates of the future spatiotemporal evolution of the main Antarctic freshwater sources in a warming climate. It can be used to force climate models that do not include interactive ice sheets, improving their representation of freshwater export and thus representing a significant advancement in accounting for the missing freshwater forcing associated with future Antarctic ice loss in climate projections.

## 2. Methods

Our Antarctic freshwater flux projections are derived from a 240-member ensemble of historically calibrated standalone ice-sheet model projections, comprehensively exploring a range of uncertain processes and parameters (Coulon, Klose, et al., 2024, see Table S1 in Supporting Information S1). All simulations were generated with the Kori-ULB ice flow model (Coulon, Klose, et al., 2024; Kazmierczak et al., 2024), which is based on the former f. ETISh model (Pattyn, 2017). The 1990–2300 pan-Antarctic simulations are forced by atmospheric and oceanic projections inferred from a subset of models from the sixth phase of the Coupled Model Intercomparison Project (CMIP6) under the low- and very high-emission scenarios SSP1-2.6 and SSP5-8.5. Although more climate scenarios are available within CMIP6, only a few models and scenarios are available until 2300.

To resolve sub-shelf processes and generate basal melt rates in such a standalone framework, we rely on a panel of physically based parameterizations that approximate the local thermal forcing based on far-field ocean properties provided by the General Circulation Models (GCMs; Jourdain et al., 2020; Favier et al., 2019; Reese et al., 2018; Lazeroms et al., 2019, Table S1 in Supporting Information S1). Similarly, freshwater discharge through iceberg calving is parameterized using semi-empirical calving laws (Morlighem et al., 2016; Pollard et al., 2015; Wilner et al., 2023). These calving laws, and in particular crevasse-depth laws, do not always produce realistically looking ice fronts and more work on iceberg calving is therefore needed. Finally, surface melt and runoff are determined within the ice-sheet model as a function of monthly air temperatures and precipitations from GCMs, using a positive degree-day (PDD)-based melt-and-runoff model to capture the basic physical processes of

refreezing versus runoff in the snow column (Coulon, Klose, et al., 2024, see Supporting Information S1). PDDbased projections have shown good agreement with climate models in terms of both pattern and magnitudes (Coulon, Klose, et al., 2024). The melt–elevation feedback is incorporated through the use of a lapse rate correction of the air temperatures with changes in ice-sheet surface elevation. In turn, near-surface air temperatures alter the amount of rainfall as well as the surface meltwater production and refreezing, so that the evolving ice-sheet topography dynamically alters the surface mass balance. Estimates of subglacial meltwater discharge across the grounding line and of blowing snow are not included here. Initially, the simulated rate of subglacial meltwater production underneath the grounded ice sheet is around 50 Gt yr<sup>-1</sup>, which is comparable to Pattyn (2010). This rate increases to up to 200 Gt yr<sup>-1</sup> during the model runs. While this amount is not negligible compared to the other freshwater fluxes, it remains difficult to constrain this as the amount of subglacial meltwater that reaches the sub-shelf cavities is unknown.

Our ice-sheet simulations start in 1990 to allow for comparisons with observations over the satellite era (Davison et al., 2023; Otosaka et al., 2023). Initial ice-sheet conditions are provided by a spin-up simulation nudging toward present-day ice-sheet geometry (Bernales et al., 2017; Coulon, Klose, et al., 2024; Pollard & DeConto, 2012a) from Bedmachine v3 (Morlighem, 2022). A historical simulation over the period 1990–2014 is then produced using surface mass balance and air temperature conditions based on the Regional Atmospheric Climate MOdel (RAC-MO2.3p2; van Wessem et al., 2018) and a present-day climatology of ocean temperature and salinity fields of the coastal ocean around Antarctica taken from Jourdain et al. (2020). This two-step approach allows us to construct a present-day ice-sheet state that reproduces both the observed geometry and changes in mass balance. The historical calibration then consists of a Bayesian calibration of our ensemble, which results in the most likely (posterior) distribution of input parameters given measured mass balance estimates and associated errors. Here we use regional mass balance estimates from the latest Ice Sheet Mass Balance Inter-comparison Exercise (IMBIE; Otosaka et al., 2023, see Supporting Information S1 and Table S2 in Supporting Information S1), which is an independent data set not used in the spin-up process. This allows a higher predictive weight to be attributed to model simulations that closely reproduce observations (Coulon, Klose, et al., 2024; Nias et al., 2019; Wernecke et al., 2020).

As of 2015, changes in atmospheric and oceanic properties derived from four CMIP6 climate models (MRI-ESM2-0, IPSL-CM6A-LR, CESM2-WACCM, and UKESM1-0-LL) are used as forcing until the year 2300. The forcing applied is derived from the Shared Socioeconomic Pathway (SSP) 1–2.6 and 5–8.5 scenarios to estimate the possible evolution of Antarctic freshwater fluxes in response to a wide range of climate forcings. Similar to Coulon, Klose, et al. (2024); Klose et al. (2024), climate forcing is implemented by applying atmospheric and oceanic anomalies to present-day climatologies, and detailed in Supporting Information S1.

To account for parametric uncertainty in our projections, we designed a perturbed parameter ensemble including 12 uncertain model parameters (see Table S1 in Supporting Information S1) that govern ice dynamics (e.g., calving, basal sliding, ice-shelf damage), ice-climate interactions (e.g., sub-shelf melting, surface melt, and runoff) and climate forcing. We adopted a two-step approach using Latin hypercube sampling to generate a comprehensive ensemble of 240 simulations. First, we generated a 120-member ensemble by uniformly sampling the multi-dimensional parameter space. Based on the resulting posterior distributions, we refined the parameter space to encompass the highest probability densities, adjusted to cover 75% cumulative probability. The posterior distributions of the parameter space for the final 240-member ensemble are displayed in Figure S1 in Supporting Information S1.

## 3. Freshwater Fluxes Projections

We provide historically calibrated yearly estimates of freshwater export from the Antarctic ice sheet from 1990 to 2300 separated into 27 drainage basins, with quantified uncertainties. Here, we analyze the results by aggregating each flux component over 5 regions of interest (Weddell Sea, Indian Ocean, western Pacific Ocean, Ross Sea and Amundsen/Bellingshausen Sea; Figure 1). However, the data for the 27 basins is available (Coulon, De Rydt, et al., 2024a).

#### 3.1. Hindcasts Over the Observational Period

The 1990–2020 average of the calibrated median total freshwater discharge from the Antarctic ice sheet is  $\sim$ 3140 Gt yr<sup>-1</sup> (Table 1). If the ice sheet were in steady state, this freshwater discharge should be balanced by the total



# **Geophysical Research Letters**

#### 10.1029/2024GL111250



**Figure 1.** Modeled Antarctic freshwater fluxes over the historical period (1990–2020). (a–j) Evolution of the regional sub-shelf melt (a–e) and calving (f–j) fluxes over the historical period and comparison with observational estimates from Davison et al. (2023) (red lines - shaded areas represent the uncertainty of the observations, shown as  $\pm$  1.64  $\sigma$ , with  $\sigma$  the observational error) for the Weddell Sea (WS), Indian Ocean (IO), Western Pacific Ocean (WPO), Ross Sea (RS) and Amundsen and Bellingshausen Sea (ABS) sectors. Dashed lines and pale blue-shaded areas represent the ensemble prior distributions (medians and 5%–95% probability intervals) while solid lines and dark blue-shaded areas represent the posterior (Bayesian calibrated medians and 5%–95% probability intervals) distributions. (k) Bayesian calibrated median regional freshwater fluxes averaged over the 1990–2020 period overlain on the 1990–2020 ensemble calibrated mean elevation rate. The boundaries of the 27 drainage basins and the 5 sectors discussed in this study are shown in gray and black, respectively.

surface accumulation. A difference between both leads to an ice sheet imbalance, with a higher discharge leading to ice mass loss. According to our model estimates, this loss is about six times the average Antarctic contribution to global mean sea-level rise over the observational period as reported in Otosaka et al. (2023), showing that much of the ice and water discharge is not contributing to sea level rise. The dominant contributor to this total mass loss is solid ice discharge from iceberg calving (59%), while liquid discharge through sub-shelf melting accounts for 40% of the total freshwater export. The modeled present-day pattern of sub-shelf melt rates is shown in Figure S2 of Supporting Information S1. In contrast, surface runoff remains a negligible contributor, accounting for less than 1% of the freshwater entering the Southern Ocean, consistent with estimates from regional climate models (Figure S3 in Supporting Information S1). This partitioning of the pan-Antarctic freshwater export aligns well with satellite-based estimates (Davison et al., 2023).



#### Table 1

Antarctic Freshwater Fluxes (Gt yr<sup>-1</sup>) Estimates Under Low- and Very-High Emission Scenarios

	Time periods				
	1990–2020	2040-2050	2090-2100	2190-2200	2290-2300
Observations (Davis	son et al., 2023)				
Sub-shelf melt	1086.2 [711.8, 1460.6]				
Iceberg calving	1542.0 [638.6, 2445.5]				
Scenario SSP1-2.6					
Total	3139.0 [1312.0,5359.0]	3767.0 [1209.3,6468.1]	3513.6 [1128.9,7243.7]	3265.9 [1055.9,8372.1]	3869.8 [939.7,9551.6]
Sub-shelf melt	1242.4 [639.0,1878.1]	1866.6 [908.9,3061.8]	1888.9 [964.3,4046.1]	1780.0 [977.0,4651.2]	2360.0 [946.7,5240.1]
Iceberg calving	1873.1 [664.6,3413.0]	1843.5 [282.3,3203.3]	1548.6 [140.0,2864.5]	1408.7 [55.0,3056.5]	1398.0 [0,3230.5]
Surface runoff	23.5 [8.5,67.9]	57.0 [18.2,202.9]	76.0 [24.5,333.1]	77.2 [24.0,664.5]	111.9 [23.5,1081.0]
Scenario SSP5-8.5					
Total	3140.6 [1312.9,5373.3]	4008.0 [1322.9,6876.0]	6273.0 [2296.6,12,630.6]	12,650.7 [5994.2,28,995.2]	14,248.8 [5635.0,29,670.1]
Sub-shelf melt	1250.5 [643.7,1905.2]	2061.9 [1036.1,3400.4]	4048.9 [1924.7,7654.9]	6991.9 [3657.0,15,674.8]	6348.8 [2753.6,10,428.4]
Iceberg calving	1865.3 [660.6,3390.2]	1842.7 [250.3,3155.1]	1532.2 [157.3,3.47.0]	2006.5 [303.3,6030.7]	2175.3 [358.8,5534.5]
Surface runoff	24.8 [8.7,77.9]	103.4 [36.5,320.5]	691.9 [214.6,1928.8]	3652.2 [2033.9,7289.7]	5724.8 [2522.6,13,707.1]

*Note.* Bayesian calibrated medians of the total Antarctic meltwater discharge and the respective contributions of sub-shelf melting, iceberg calving, and surface meltwater runoff for the 1990–2020, 2090–2100, 2190–2200, and 2290–2300 time periods under SSP1-2.6 and SSP5-8.5. Values in brackets for the scenarios SSP1-2.6 and SSP5-8.5 are the Bayesian calibrated 5th–95th percentiles (likely range). Values in brackets for the observations are shown as the estimates  $\pm$  1.64  $\sigma$ , with  $\sigma$  the observational error.

According to our model results, the majority of the pan-Antarctic freshwater discharge over 1990–2020 occurs in the Weddell and Amundsen/Bellingshausen Seas (Figure 1). Iceberg calving is the dominant contributor to freshwater discharge in all regions except the Amundsen/Bellingshausen Sea, where the Bayesian calibrated median sub-shelf melt contributes to 62% of the total freshwater flux. Runoff hindcasts are negligible in all 5 regions (with the highest median contribution of 2% in the Amundsen/Bellingshausen Sea, originating from the Antarctic Peninsula; see Figure S3 in Supporting Information S1). The partitioning of freshwater flux components for different regions is also consistent with observational estimates (Davison et al., 2023; Figures 1a–1j).

When equal weight is assigned to all simulations, our ensemble displays a large spread in freshwater fluxes over the historical period due to the sampled parametric uncertainty (Figures 1a–1j). However, attributing a lower weight to parameter choices that are not in line with observations through the Bayesian calibration reduces the spread of the posterior distribution compared to the prior, thereby reducing uncertainty in the model hindcasts and increasing confidence in the projections through 2300. In particular, the posterior distributions of parameters related to sub-shelf melting, iceberg calving, and ice-shelf damage are most affected by the calibration process (Figure S1 in Supporting Information S1). Similarly, our modeled spatial pattern of mass loss over the historical period (Figure 1k) shows rapid grounded-ice mass losses around the margins of the ice sheet, particularly in the Amundsen Sea Embayment and the Aurora Basin in East Antarctica, in line with observations over the past decades (Smith et al., 2020; Rignot et al., 2019; Nilsson et al., 2022, see Figure S4 in Supporting Information S1).

#### 3.2. Projections to 2300

By 2100, we project an increase in total pan-Antarctic freshwater fluxes of 12% under SSP1-2.6, and of 100% under SSP5-8.5 (Table 1), mostly in the Amundsen/Bellingshausen and Weddell Seas (Figure 2). Sub-shelf melt (and to a lesser extent surface melt) increases under both scenarios during the first half of this century, leading to a slight decrease in iceberg calving fluxes. Consequently, sub-shelf melting is projected to become the dominant freshwater contributor by 2100, surpassing calving and shifting the balance from predominantly solid to liquid freshwater export. This transition is expected to occur by around 2050 under both scenarios. Meltwater runoff remains negligible under SSP1-2.6 but starts to increase in the second half of the century under SSP5-8.5, reaching about 10% of the total freshwater discharge by the end of the century.



# **Geophysical Research Letters**



**Figure 2.** Calibrated probabilistic projections of the Antarctic freshwater fluxes until the year 2300. (a–b) Evolution of the ensemble projected pan-Antarctic freshwater fluxes (sub-shelf melt, calving, and runoff) for the 1990–2300 period under SSP1-2.6 (a) and SSP5-8.5 (b). Solid lines and shaded regions show the medians and 5%–95% probability intervals (N = 240 per SSP scenario), with 5-year running average applied. Boxes and whiskers show [5, 25, 50, 75, 95] percentiles at year 2300. (c) Evolution of the ensemble projected regional freshwater fluxes for the 1990–2300 period under an SSP5-8.5 scenario overlain on the 1990–2300 ensemble calibrated mean elevation rate. Note the differing scales on the *y*-axes. Pie charts represent the median regional freshwater fluxes averaged over the 2090–2100 and the 2290–2300 periods.

While sub-shelf melt fluxes are predicted to plateau in the second half of this century under the low-emission scenario, they keep increasing under SSP5-8.5, reaching twice their 2040–2050 magnitude by 2100 (Table 1). This difference in the amplitude of sub-shelf melting between the two emission scenarios is especially pronounced in the Weddell Sea, where fluxes under SSP5-8.5 reach four times those of SSP1-2.6 by 2100 (see Figure S5 in Supporting Information S1). This is likely due to the Ronne-Filchner ice shelf transitioning from a cold to a warm cavity (Hellmer et al., 2017; Naughten et al., 2021). In addition, the unique partitioning of the freshwater sources in the Amundsen/Bellingshausen Sea under SSP5-8.5 is noteworthy (Figure 2). Unlike other regions, by 2100, freshwater sources in this sector are characterized by a substantial increase in surface runoff, representing almost 30% of the total (Bayesian calibrated median), and mainly originating from the Antarctic Peninsula side of the Bellingshausen Sea. In the other sectors, meltwater runoff remains the smallest median contributor to the freshwater discharge, never exceeding 10%.



# **Geophysical Research Letters**



**Figure 3.** Projected changes in Antarctic geometry until 2300 under low- and very-high emission scenarios. (a–b) Mean ice thickness change in years 2050, 2100, 2200, and 2300 under the shared socio-economic pathways (SSP) 1–2.6 (a) and 5–8.5 (b). The mean thickness change at a given point in time is computed using the Bayesian calibrated mean of the ensemble (N = 240 per SSP scenario). Black and gray lines show the ensemble mean grounding line and calving front positions, respectively. (c–d) Bayesian calibrated mean depth profile of freshwater discharge through sub-shelf melting at years 2000, 2100, 2200 and 2300, for the Weddell Sea (WS), Indian Ocean (IO), Western Pacific Ocean (WPO), Ross Sea (RS) and Amundsen and Bellingshausen Sea (ABS) sectors under SSP1-2.6 (c) and 5–8.5 (d). Dashed and dotted lines show the evolution of the Bayesian calibrated mean ice front draft and grounding-line depth for each region, respectively.

After 2100, the pattern of projected Antarctic ice discharge starts to significantly diverge under the two emission scenarios (Figures 3a and 3b). Under SSP1-2.6, mass loss remains largely confined to the Amundsen Sea Embayment. In contrast, under SSP5-8.5, there is an acceleration of mass loss linked to a progressive thinning and collapse of the West Antarctic ice shelves, leading to rapid grounded ice loss. In addition, we note the onset of

grounding-line retreat in the marine basins of East Antarctica, despite a thickening of a few tens of meters of the inland ice sheet.

As a consequence, by 2300, the median total pan-Antarctic freshwater flux increases by 23% under SSP1-2.6 and by 354% under SSP5-8.5 compared to the 1990–2020 average (Figure 2 and Table 1). Under the low-emission scenario, the partitioning of the freshwater discharge remains overall unchanged after 2100 (Table 1), both at the continental and regional scales (Figure S5 in Supporting Information S1). In contrast, under SSP5-8.5, the increase in sub-shelf melting triggered in the second half of the 21st century continues beyond 2100. Sub-shelf melt fluxes stabilize in the second half of the 22nd century, reaching nearly 7000 Gt yr<sup>-1</sup> by 2200, and then eventually decline with the ongoing collapse of the ice shelves (Figures 2b and 3b). Surface runoff increases steadily with warming air temperatures, reaching median fluxes of about 6000 Gt yr<sup>-1</sup> by 2300 (i.e., of the same order of magnitude as the sub-shelf melt), and potentially becoming the dominant source of freshwater discharge (5%-95% probability interval; Figure 2b and Table 1). Regionally, median surface runoff fluxes dominate the freshwater discharge in the Indian Ocean by 2300, and are of a similar magnitude to sub-shelf melting in the other regions. Unlike the other two freshwater sources, calving fluxes under SSP5-8.5 remain fairly constant after 2100. They may slightly increase after 2150, likely due to the decrease in total sub-shelf melting, which is caused by an overall reduction in ice-shelf area (Figure 2), particularly within the 5%-95% probability interval. Overall, consistent with the 21st century trend, most of the freshwater will be discharged into the Weddell, Amundsen, and Bellingshausen Seas by 2300 (Figures 2 and 3).

#### 4. Discussion

We present spatially integrated timeseries of freshwater fluxes for the main Antarctic drainage basins, here aggregated over 5 oceanic sectors. Therefore, our estimates do not consider details in both horizontal and vertical dimensions, which may influence climate dynamics (Pauling et al., 2016; Thomas et al., 2023). Although our model simulations contain information about the horizontal distribution of individual freshwater sources down to the model grid scale (8-km resolution), we do not include it in this analysis. Given the often coarse resolution of CMIP-class climate models, the spatial distribution of freshwater fluxes at kilometer scales is unlikely to significantly affect their response, as these sources will rapidly merge and diffuse away from the coastline. In addition, while the regional distribution of freshwater fluxes is key to understanding local ice-sheet–climate feedbacks (such as changes in ocean dynamics and melt rates due to changes in upstream freshwater release), our standalone ice-sheet simulations forced by CMIP6 data do not capture such feedbacks. We therefore focus on the basin-scale trends and partitioning of different freshwater sources for a range of uncertain ice-sheet processes.

In the case of sub-shelf melting, changes in the draft of the ice shelves can cause a shift in the vertical distribution of freshwater input. The level at which freshwater is injected into the ocean is a key input for simulating the basalmelt-induced overturning circulation and associated changes in ocean stratification on the continental shelves (Mathiot et al., 2017; Thomas et al., 2023). Although we do not provide a comprehensive analysis here, we show that increasing meltwater is projected to be discharged at greater depths through sub-shelf melting in the Amundsen, Bellinghsausen and Ross Seas as well as in the western Pacific Ocean over time, due to the projected grounding-line retreat in the main Antarctic marine basins (Figure 3).

Our projections also do not simulate abrupt discrete calving events or provide detailed spatial distribution of calving fluxes at the ice front, complicating the estimation of where and when freshwater will be released at the ocean's surface. Nonetheless, the ice-shelf thinning projected by our calibrated ensemble indicates an overall decrease in the ice front draft and hence iceberg thickness in the coming centuries (Figure 3 and Figure S6 in Supporting Information S1). Future work should focus on providing more detailed maps of future freshwater fluxes. Meanwhile, our basin-integrated projections can be added as anomalies to present-day spatial distributions (Thomas et al., 2023) derived from observations (e.g., Rignot et al., 2013), allowing for the consideration of regional climate feedback sensitivities (Gorte et al., 2023; Thomas et al., 2023).

In addition, our projections lack a surface meltwater routing and storage scheme. We hence provide a first-order estimate by assuming that all surface meltwater not refrozen in winter is routed laterally toward the ocean within each drainage basin. In reality, ponding of meltwater has been widely observed in Antarctica, especially on the flat ice shelves (Kingslake et al., 2017; van Wessem et al., 2023). Meltwater storage has also been observed in the form of englacial lakes (Lenaerts et al., 2016) and perennial firm aquifers (van Wessem et al., 2021).

19448007, 2024, 23, Downloaded from https:

elibrary.wiley.com/doi/10.1029/2024GL111250 by Université De Liège, Wiley Online Library on [09/12/2024]. See the Terms

(nup

morary, wiley

of use; OA articles

by the applicable Creat

Despite accounting for a comprehensive range of uncertain processes, our projections of future freshwater fluxes from the Antarctic ice sheet rely on a single ice-sheet model, hence lacking structural uncertainty. Our estimates are also influenced by the limited number of climate models providing projections until 2300 and the emission scenarios for which such projections are produced (here, only the low and very high emission scenarios). An assessment of the sensitivity of future Antarctic freshwater fluxes to intermediate pathways is thus lacking. Moreover, the SSP5-8.5 scenario is considered very unlikely to be followed on multi-centennial timescales (Hausfather & Peters, 2020; Schwalm et al., 2020). Our projections under this very-high emission scenario may therefore be seen as high-end estimates. While calibration adds robustness to projections, it is important to recognize that exploring additional sources of uncertainties will likely alter and broaden the distributions of the projections (Edwards et al., 2021). Similarly, alternative choices for the magnitude of the structural error (and therefore discrepancy variance; see Supporting Information S1) would also influence our calibrated projections (Coulon, Klose, et al., 2024). A future comparison with freshwater fluxes from the ISMIP6 extended ensemble (Seroussi et al., 2024) could aid in limiting the structural uncertainty.

#### 5. Conclusion

Large changes in the spatial distribution, magnitude and partitioning (calving vs. ice-shelf melt vs. surface runoff) of future Antarctic freshwater discharge are expected in the coming decades and centuries, particularly under high-emission scenarios. Overall, the freshwater release from the Antarctic ice sheet into the Southern Ocean, currently amounting to about 3140 Gt yr<sup>-1</sup>, is expected to increase. While this increase remains limited under a low emission scenario, it reaches a factor of 2 (with almost 6300 Gt yr<sup>-1</sup>) by 2100, and 4.5 ( $\sim$ 14,250 Gt yr<sup>-1</sup>) by 2300 under a very-high emission scenario, significantly increasing the potential to trigger feedbacks within the global climate system. We also project a shift in the form and depth of freshwater discharge from solid to liquid, as the dominant contributor to Antarctic freshwater fluxes is expected to transition from calving to sub-shelf melt, and potentially to surface meltwater runoff, which may become a dominant contributor under a very-high emission scenario. Although almost all Antarctic regions are projected to experience increased freshwater discharge, most of this discharge will occur in the Amundsen and Weddell Seas. Given that some climate model studies have already found significant climate impacts from freshwater forcing changes of only a few percent (e.g., Bintanja et al., 2013), accounting for these large spatiotemporal trends in climate projections will allow for a better assessment and quantification of the climate system's response and feedback to changing ice-sheet-driven freshwater input from Antarctica, as well as the role of uncertain ice-sheet processes such as calving in that response.

### **Data Availability Statement**

The code and reference manual of the Kori-ULB ice-sheet model are publicly available on Github and can be downloaded from Pattyn (2024). The code of the specific Kori-ULB model version used in this study is currently available on Github from Coulon (2024). This version of the code is hosted on Zenodo, along with the simulation outputs and the scripts needed to produce the figures and tables in this paper (https://doi.org/10.5281/zenodo. 14163522, Coulon, De Rydt, et al., 2024b). The freshwater flux estimates are also available on Zenodo (https://doi.org/10.5281/zenodo. 14163522, Coulon, De Rydt, et al., 2024b). The freshwater flux estimates are also available on Zenodo (https://doi.org/10.5281/zenodo.14162775, Coulon, De Rydt, et al., 2024a). All data sets used in this study are freely accessible through their original references. The CMIP6 forcing data used in this study are accessible through the CEDA ESGF node (CEDA ESGF, 2024). Ice and bedrock geometry data from Bedmachine v3 are available in Morlighem (2022). Outputs from RACMOv2.3p2 are available in van Wessem et al. (2018). Present-day ocean climatology data are available in Jourdain et al. (2020). Ice-sheet mass balance estimates are available in Otosaka et al. (2023) and calving and sub-shelf melt fluxes estimates are available in Davison et al. (2023).

### References

Adusumilli, S., Fricker, H., Medley, B., Padman, L., & Siegfried, M. (2020). Interannual variations in meltwater input to the southern ocean from Antarctic ice shelves. *Nature Geoscience*, 13, 1–5. https://doi.org/10.1038/s41561-020-0616-z

Bintanja, R., van Oldenborgh, G. J., Drijfhout, S. S., Wouters, B., & Katsman, C. A. (2013). Important role for ocean warming and increased iceshelf melt in Antarctic sea-ice expansion. *Nature Geoscience*, 6(5), 376–379. https://doi.org/10.1038/ngeo1767

## Acknowledgments

This research was supported by OCEAN: ICE, which is co-funded by the European Union, Horizon Europe Funding Programme for research and innovation under grant agreement Nr. 101060452 and by UK Research and Innovation. OCEAN: ICE Contribution number 11. Computational resources have been provided by the Consortium des Équipements de Calcul Intensif (CÉCI), funded by the Fonds de la Recherche Scientifique de Belgique (F.R.S.-FNRS) under Grant Number 2.5020.11 and by the Walloon Region. TG was supported by the F.R.S.-FNRS through a Research Fellowship. JDR was also supported by a UKRI Future Leaders Fellowship under Grant number MR/W011816/1

Bell, R. E., Chu, W., Kingslake, J., Das, I., Tedesco, M., Tinto, K. J., et al. (2017). Antarctic ice shelf potentially stabilized by export of meltwater in surface river. *Nature*, 544(7650), 344–348. https://doi.org/10.1038/nature22048

Bernales, J., Rogozhina, I., & Thomas, M. (2017). Melting and freezing under Antarctic ice shelves from a combination of ice-sheet modelling and observations. *Journal of Glaciology*, 63(240), 731–744. https://doi.org/10.1017/jog.2017.42

Bronselaer, B., Winton, M., Griffies, S., Hurlin, W., Rodgers, K., Sergienko, O., et al. (2018). Change in future climate due to Antarctic meltwater. *Nature*, 564(7734), 53–58. https://doi.org/10.1038/s41586-018-0712-z

CEDA, ESGF. (2024). Wcrp coupled model intercomparison project (phase 6) [Dataset]. https://esgf-ui.ceda.ac.uk/cog/projects/cmip6-ceda Coulon, V. (2024). Kori-ULB [Software]. *GitHub repository*. https://github.com/VioCoulon/Kori-ULB/tree/OceanIce

Coulon, V., De Rydt, J., Gregov, T., Qin, Q., & Pattyn, F. (2024a). Future freshwater fluxes from the Antarctic ice sheet [Dataset]. Zenodo. https:// doi.org/10.5281/zenodo.14162775

- Coulon, V., Klose, A. K., Kittel, C., Edwards, T., Turner, F., Winkelmann, R., & Pattyn, F. (2024). Disentangling the drivers of future Antarctic ice loss with a historically calibrated ice-sheet model. *The Cryosphere*, 18(2), 653–681. https://doi.org/10.5194/tc-18-653-2024
- Davison, B. J., Hogg, A. E., Gourmelen, N., Jakob, L., Wuite, J., Nagler, T., et al. (2023). Annual mass budget of Antarctic ice shelves from 1997 to 2021. Science Advances, 9(41), eadi0186. https://doi.org/10.1126/sciadv.adi0186
- de Lavergne, C., Palter, J. B., Galbraith, E. D., Bernardello, R., & Marinov, I. (2014). Cessation of deep convection in the open southern ocean under anthropogenic climate change. *Nature Climate Change*, 4(4), 278–282. https://doi.org/10.1038/nclimate2132
- Depoorter, M. A., Bamber, J. L., Griggs, J. A., Lenaerts, J. T. M., Ligtenberg, S. R. M., van den Broeke, M. R., & Moholdt, G. (2013). Calving fluxes and basal melt rates of Antarctic ice shelves. *Nature*, 502(7469), 89–92. https://doi.org/10.1038/nature12567
- Edwards, T. L., Nowicki, S., Marzeion, B., Hock, R., Goelzer, H., Seroussi, H., et al. (2021). Projected land ice contributions to 21st century sea level rise. *Nature*, 593(7857), 74–82. https://doi.org/10.1038/s41586-021-03302-y
- Favier, L., Jourdain, N. C., Jenkins, A., Merino, N., Durand, G., Gagliardini, O., et al. (2019). Assessment of sub-shelf melting parameterisations using the ocean-ice-sheet coupled model nemo(v3.6)-elmer/ice(v8.3). *Geoscientific Model Development*, 12(6), 2255–2283. https://doi.org/ 10.5194/gmd-12-2255-2019
- Fox-Kemper, B., Hewitt, H., Xiao, C., Aeth algeirsdóttir, G., Drijfhout, S., Edwards, T., et al. (2021). Ocean, cryosphere and sea level change [Book Section]. In Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 1211–1362). Cambridge University Press. https://doi.org/10.1017/9781009157896.011
- Golledge, N., Keller, E., Gomez, N., Naughten, K., Bernales, J., Trusel, L., & Edwards, T. (2019). Global environmental consequences of twentyfirst-century ice-sheet melt. *Nature*, 566(7742), 65–72. https://doi.org/10.1038/s41586-019-0889-9
- Gorte, T., Lovenduski, N. S., Nissen, C., & Lenaerts, J. T. M. (2023). Antarctic ice sheet freshwater discharge drives substantial southern ocean changes over the 21st century. *Geophysical Research Letters*, 50(20), e2023GL104949. https://doi.org/10.1029/2023GL104949
- Greene, C. A., Gardner, A. S., Schlegel, N.-J., & Fraser, A. D. (2022). Antarctic calving loss rivals ice-shelf thinning. *Nature*, 609(7929), 948–953. https://doi.org/10.1038/s41586-022-05037-w
- Hausfather, Z., & Peters, G. (2020). 01). Emissions The 'business as usual' story is misleading. *Nature*, 577(7792), 618–620. https://doi.org/10. 1038/d41586-020-00177-3
- Hellmer, H. H., Kauker, F., Timmermann, R., & Hattermann, T. (2017). The fate of the southern Weddell Sea continental shelf in a warming climate. Journal of Climate, 30(12), 4337–4350. https://doi.org/10.1175/JCLI-D-16-0420.1
- Jourdain, N. C., Asay-Davis, X., Hattermann, T., Straneo, F., Seroussi, H., Little, C. M., & Nowicki, S. (2020). A protocol for calculating basal melt rates in the ismip6 Antarctic ice sheet projections. *The Cryosphere*, 14(9), 3111–3134. https://doi.org/10.5194/tc-14-3111-2020
- Kazmierczak, E., Gregov, T., Coulon, V., & Pattyn, F. (2024). A fast and unified subglacial hydrological model applied to Thwaite's glacier, Antarctica. EGUsphere, 2024, 1–36. https://doi.org/10.5194/egusphere-2024-466
- Kingslake, J., Ely, J. C., Das, I., & Bell, R. E. (2017). Widespread movement of meltwater onto and across Antarctic ice shelves. Nature, 544(7650), 349–352. https://doi.org/10.1038/nature22049
- Klose, A. K., Coulon, V., Pattyn, F., & Winkelmann, R. (2024). The long-term sea-level commitment from Antarctica. *The Cryosphere*, 18(9), 4463–4492. https://doi.org/10.5194/tc-18-4463-2024
- Lawrence, I. R., Ridout, A. L., Shepherd, A., & Tilling, R. (2023). A simulation of snow on Antarctic sea ice based on satellite data and climate reanalyses. *Journal of Geophysical Research: Oceans*, 129(1), e2022JC019002. https://doi.org/10.1029/2022jc019002
- Lazeroms, W. M. J., Jenkins, A., Rienstra, S. W., & van de Wal, R. S. W. (2019). An analytical derivation of ice-shelf basal melt based on the dynamics of meltwater plumes. *Journal of Physical Oceanography*, 49(4), 917–939. https://doi.org/10.1175/JPO-D-18-0131.1
- Lenaerts, J., Vizcaíno, M., Fyke, J., van Kampenhout, L., & Van den Broeke, M. (2016). Present-day and future Antarctic ice sheet climate and surface mass balance in the Community Earth System Model. *Climate Dynamics*, 47(5–6), 1367–1381. https://doi.org/10.1007/s00382-015-2907-4
- Mathiot, P., Jenkins, A., Harris, C., & Madec, G. (2017). Explicit representation and parametrised impacts of under ice shelf seas in the z<sup>ast</sup> coordinate ocean model nemo 3.6. *Geoscientific Model Development*, 10(7), 2849–2874. https://doi.org/10.5194/gmd-10-2849-2017
- Morlighem, M. (2022). Measures bed machine Antarctica, version 3. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/FPSU0V1MWUB6
- Morlighem, M., Bondzio, J., Seroussi, H., Rignot, E., Larour, E., Humbert, A., & Rebuffi, S. (2016). Modeling of store gletscher's calving dynamics, west Greenland, in response to ocean thermal forcing. *Geophysical Research Letters*, 43(6), 2659–2666. https://doi.org/10.1002/ 2016GL067695
- Mottram, R., Hansen, N., Kittel, C., van Wessem, J. M., Agosta, C., Amory, C., et al. (2021). What is the surface mass balance of Antarctica? An intercomparison of regional climate model estimates. *The Cryosphere*, *15*(8), 3751–3784. https://doi.org/10.5194/tc-15-3751-2021
- Naughten, K. A., De Rydt, J., Rosier, S. H. R., Jenkins, A., Holland, P. R., & Ridley, J. K. (2021). Two-timescale response of a large Antarctic ice shelf to climate change. *Nature Communications*, 12(1), 1991. https://doi.org/10.1038/s41467-021-22259-0
- Nias, I. J., Cornford, S. L., Edwards, T. L., Gourmelen, N., & Payne, A. J. (2019). Assessing uncertainty in the dynamical ice response to ocean warming in the Amundsen Sea embayment, West Antarctica. *Geophysical Research Letters*, 46(20), 11253–11260. https://doi.org/10.1029/ 2019GL084941
- Nilsson, J., Gardner, A. S., & Paolo, F. S. (2022). Elevation change of the Antarctic ice sheet: 1985 to 2020. Earth System Science Data, 14(8), 3573–3598. https://doi.org/10.5194/essd-14-3573-2022
- Otosaka, I. N., Shepherd, A., Ivins, E. R., Schlegel, N.-J., Amory, C., van den Broeke, M. R., et al. (2023). Mass balance of the Greenland and Antarctic ice sheets from 1992 to 2020. *Earth System Science Data*, *15*(4), 1597–1616. https://doi.org/10.5194/essd-15-1597-2023
- Paolo, F. S., Gardner, A. S., Greene, C. A., Nilsson, J., Schodlok, M. P., Schlegel, N.-J., & Fricker, H. A. (2023). Widespread slowdown in thinning rates of west Antarctic ice shelves. *The Cryosphere*, 17(8), 3409–3433. https://doi.org/10.5194/tc-17-3409-2023
- Pattyn, F. (2010). Antarctic subglacial conditions inferred from a hybrid ice sheet/ice stream model. *Earth and Planetary Science Letters*, 295(3), 451–461. https://doi.org/10.1016/j.epsl.2010.04.025

Coulon, V., De Rydt, J., Gregov, T., Qin, Q., & Pattyn, F. (2024b). Future freshwater fluxes from the Antarctic ice sheet [Software]. Zenodo. https://doi.org/10.5281/zenodo.14163522

Pattyn, F. (2017). Sea-level response to melting of Antarctic ice shelves on multi-centennial timescales with the fast Elementary Thermomechanical Ice Sheet model (f.ETISh v1.0). *The Cryosphere*, *11*(4), 1–28. https://doi.org/10.5194/tc-11-1-2017

Pattyn, F. (2024). Kori-ULB [Software]. GitHub repository. https://github.com/FrankPat/Kori-dev

Pauling, A. G., Bitz, C. M., Smith, I. J., & Langhorne, P. J. (2016). The response of the southern ocean and Antarctic sea ice to freshwater from ice shelves in an earth system model. *Journal of Climate*, 29(5), 1655–1672. https://doi.org/10.1175/JCLI-D-15-0501.1

- Payne, A., Nowicki, S., Abe-Ouchi, A., Agosta, C., Alexander, P., Albrecht, T., et al. (2021). 08). Future sea level change under cmip5 and cmip6 scenarios from the Greenland and Antarctic ice sheets. *Geophysical Research Letters*, 48(16), e2020GL091741. https://doi.org/10.1029/ 2020GL091741
- Pelletier, C., Fichefet, T., Goosse, H., Haubner, K., Helsen, S., Huot, P.-V., et al. (2022). PARASO, a circum-Antarctic fully coupled ice-sheet– ocean–sea-ice–atmosphere–land model involving f.ETISh1.7, NEMO3.6, LIM3.6, COSMO5.0 and CLM4.5. Geoscientific Model Development, 15(2), 553–594. https://doi.org/10.5194/gmd-15-553-2022
- Pollard, D., & DeConto, R. M. (2012a). A simple inverse method for the distribution of basal sliding coefficients under ice sheets, applied to Antarctica. *The Cryosphere*, 6(5), 953–971. https://doi.org/10.5194/tc-6-953-2012
- Pollard, D., DeConto, R. M., & Alley, R. B. (2015). Potential Antarctic Ice Sheet retreat driven by hydrofracturing and ice cliff failure. *Earth and Planetary Science Letters*, 412, 112–121. https://doi.org/10.1016/j.epsl.2014.12.035
- Purich, A., & England, M. H. (2023). Projected impacts of antarctic meltwater anomalies over the twenty-first century. *Journal of Climate*, 36(8), 2703–2719. https://doi.org/10.1175/jcli-d-22-0457.1
- Reese, R., Albrecht, T., Mengel, M., Asay-Davis, X., & Winkelmann, R. (2018). Antarctic sub-shelf melt rates via Pico. *The Cryosphere*, 12(6), 1969–1985. https://doi.org/10.5194/tc-12-1969-2018
- Rignot, E., Jacobs, S., Mouginot, J., & Scheuchl, B. (2013). Ice-shelf melting around Antarctica. Science, 341(6143), 266–270. https://doi.org/10. 1126/science.1235798
- Rignot, E., Mouginot, J., Scheuchl, B., van den Broeke, M., van Wessem, M. J., & Morlighem, M. (2019). Four decades of Antarctic ice sheet mass balance from 1979–2017. Proceedings of the National Academy of Sciences, 116(4), 1095–1103. https://doi.org/10.1073/pnas.1812883116
- Sadai, S., Condron, A., DeConto, R., & Pollard, D. (2020). Future climate response to Antarctic Ice Sheet melt caused by anthropogenic warming. Science Advances, 6(39), eaaz1169. https://doi.org/10.1126/sciadv.aaz1169
- Schloesser, F., Friedrich, T., Timmermann, A., DeConto, R. M., & Pollard, D. (2019). Antarctic iceberg impacts on future southern hemisphere climate. *Nature Climate Change*, 9(9), 672–677. https://doi.org/10.1038/s41558-019-0546-1
- Schwalm, C. R., Glendon, S., & Duffy, P. B. (2020). RCP8.5 tracks cumulative CO2 emissions. Proceedings of the National Academy of Sciences, 117(33), 19656–19657. https://doi.org/10.1073/pnas.2007117117
- Seroussi, H., Nowicki, S., Payne, A. J., Goelzer, H., Lipscomb, W. H., Abe-Ouchi, A., et al. (2020). Ismip6 Antarctica: A multi-model ensemble of the Antarctic ice sheet evolution over the 21st century. *The Cryosphere*, 14(9), 3033–3070. https://doi.org/10.5194/tc-14-3033-2020
- Seroussi, H., Pelle, T., Lipscomb, W. H., Abe-Ouchi, A., Albrecht, T., Alvarez-Solas, J., et al. (2024). Evolution of the Antarctic ice sheet over the next three centuries from an ismip6 model ensemble. *Earth's Future*, 12(9), e2024EF004561. https://doi.org/10.1029/2024EF004561
- Siahaan, A., Smith, R. S., Holland, P. R., Jenkins, A., Gregory, J. M., Lee, V., et al. (2022). The Antarctic contribution to 21st-century sea-level rise predicted by the UK earth system model with an interactive ice sheet. *The Cryosphere*, 16(10), 4053–4086. https://doi.org/10.5194/tc-16-4053-2022
- Silvano, A., Rintoul, S. R., Peña-Molino, B., Hobbs, W. R., van Wijk, E., Aoki, S., et al. (2018). Freshening by glacial meltwater enhances melting of ice shelves and reduces formation of Antarctic bottom water. *Science Advances*, 4(4), eaap9467. https://doi.org/10.1126/sciady.aap9467
- Smith, B., Fricker, H. A., Gardner, A. S., Medley, B., Nilsson, J., Paolo, F. S., et al. (2020). Pervasive ice sheet mass loss reflects competing ocean and atmosphere processes. *Science*, 368(6496), 1239–1242. https://doi.org/10.1126/science.aaz5845
- Swart, N. C., Martin, T., Beadling, R., Chen, J.-J., Danek, C., England, M. H., et al. (2023). The Southern Ocean freshwater input from Antarctica (Sofia) initiative: Scientific objectives and experimental design. *Geoscientific Model Development*, 16(24), 7289–7309. https://doi.org/10. 5194/gmd-16-7289-2023
- Thomas, M., Ridley, J. K., Smith, I. J., Stevens, D. P., Holland, P. R., & Mackie, S. (2023). Future response of Antarctic continental shelf temperatures to ice shelf basal melting and calving. *Geophysical Research Letters*, 50(18), e2022GL102101. https://doi.org/10.1029/ 2022GL102101
- Trusel, L., Frey, K., Das, S., Karnauskas, K., Munneke, P., Meijgaard, E., & Van den Broeke, M. (2015). Divergent trajectories of Antarctic surface melt under two twenty-first-century climate scenarios. *Nature Geoscience*, 8(12), 927–932. https://doi.org/10.1038/ngeo2563
- Trusel, L., Frey, K. E., Das, S. B., Munneke, P. K., & van den Broeke, M. R. (2013). Satellite-based estimates of Antarctic surface meltwater fluxes. *Geophysical Research Letters*, 40(23), 6148–6153. https://doi.org/10.1002/2013GL058138
- van Wessem, J. M., Steger, C. R., Wever, N., & van den Broeke, M. R. (2021). An exploratory modelling study of perennial firm aquifers in the antarctic peninsula for the period 1979–2016. *The Cryosphere*, *15*(2), 695–714. https://doi.org/10.5194/tc-15-695-2021
- van Wessem, J. M., van de Berg, W. J., Noël, B. P. Y., van Meijgaard, E., Amory, C., Birnbaum, G., et al. (2018). Modelling the climate and surface mass balance of polar ice sheets using racmo2 Part 2: Antarctica (1979–2016). *The Cryosphere*, *12*(4), 1479–1498. https://doi.org/10. 5194/tc-12-1479-2018
- van Wessem, J. M., Van den Broeke, M., Wouters, B., & Lhermitte, S. (2023). Variable temperature thresholds of melt pond formation on Antarctic ice shelves. *Nature Climate Change*, 13(2), 1–6. https://doi.org/10.1038/s41558-022-01577-1
- Wernecke, A., Edwards, T. L., Nias, I. J., Holden, P. B., & Edwards, N. R. (2020). Spatial probabilistic calibration of a high-resolution Amundsen Sea embayment ice sheet model with satellite altimeter data. *The Cryosphere*, *14*(5), 1459–1474. https://doi.org/10.5194/tc-14-1459-2020
- Wilner, J. A., Morlighem, M., & Cheng, G. (2023). Evaluation of four calving laws for Antarctic ice shelves. *The Cryosphere*, 17(11), 4889–4901. https://doi.org/10.5194/tc-17-4889-2023

## **References From the Supporting Information**

- Bondzio, J. H., Seroussi, H., Morlighem, M., Kleiner, T., Rückamp, M., Humbert, A., & Larour, E. Y. (2016). Modelling calving front dynamics using a level-set method: Application to Jakobshavn isbræ, west Greenland. *The Cryosphere*, 10(2), 497–510. https://doi.org/10.5194/tc-10-497-2016
- Braithwaite, R. J. (2008). Temperature and precipitation climate at the equilibrium-line altitude of glaciers expressed by the degree-day factor for melting snow. Journal of Glaciology, 54(186), 437–444. https://doi.org/10.3189/002214308785836968
- Brotchie, J. F., & Silvester, R. (1969). On crustal flexure. Journal of Geophysical Research, 74(22), 5240-5252. https://doi.org/10.1029/ JB074i022p05240

- Bueler, E., & Brown, J. (2009). Shallow shelf approximation as a "sliding law" in a thermomechanically coupled ice sheet model. Journal of Geophysical Research, 114(F3). https://doi.org/10.1029/2008JF001179
- Burgard, C., Jourdain, N. C., Reese, R., Jenkins, A., & Mathiot, P. (2022). An assessment of basal melt parameterisations for Antarctic ice shelves. *The Cryosphere*, 16(12), 4931–4975. https://doi.org/10.5194/tc-16-4931-2022
- Coulon, V., Bulthuis, K., Whitehouse, P. L., Sun, S., Haubner, K., Zipf, L., & Pattyn, F. (2021). Contrasting response of west and east Antarctic ice sheets to glacial isostatic adjustment. *Journal of Geophysical Research: Earth Surface*, 126(7), e2020JF006003. https://doi.org/10.1029/ 2020JF006003
- Cuffey, K. M., & Paterson, W. S. B. (2010). The physics of glaciers. Academic Press.
- Edwards, T. L., Brandon, M. A., Durand, G., Edwards, N. R., Golledge, N. R., Holden, P. B., et al. (2019). Revisiting Antarctic ice loss due to marine ice-cliff instability. *Nature*, 566(7742), 58–64. https://doi.org/10.1038/s41586-019-0901-4
- Hock, R. (2003). Temperature index melt modelling in mountain areas. Journal of Hydrology, 282(1), 104–115. https://doi.org/10.1016/S0022-1694(03)00257-9
- Hock, R. (2005). Glacier melt: A review of processes and their modelling. Progress in Physical Geography: Earth and Environment, 29(3), 362– 391. https://doi.org/10.1191/0309133305pp453ra
- Huybrechts, P., & de Wolde, J. (1999). The dynamic response of the Greenland and Antarctic ice sheets to multiple-century climatic warming. Journal of Climate, 12(8), 2169–2188. https://doi.org/10.1175/1520-0442(1999)012<2169:tdrotg>2.0.co;2
- Kittel, C., Amory, C., Agosta, C., Jourdain, N. C., Hofer, S., Delhasse, A., et al. (2021). Diverging future surface mass balance between the Antarctic ice shelves and grounded ice sheet. *The Cryosphere*, 15(3), 1215–1236. https://doi.org/10.5194/tc-15-1215-2021
- Kreuzer, M., Reese, R., Huiskamp, W. N., Petri, S., Albrecht, T., Feulner, G., & Winkelmann, R. (2021). Coupling framework (1.0) for the pism (1.1.4) ice sheet model and the mom5 (5.1.0) ocean model via the pico ice shelf cavity model in an Antarctic domain. *Geoscientific Model Development*, 14(6), 3697–3714. https://doi.org/10.5194/gmd-14-3697-2021
- Le Meur, E., & Huybrechts, P. (1996). A comparison of different ways of dealing with isostasy: Examples from modelling the Antarctic ice sheet during the last glacial cycle. Annals of Glaciology, 23, 309–317. https://doi.org/10.10013/epic.12717.d001
- Magand, O., Frezzotti, M., Pourchet, M., Stenni, B., Genoni, L., & Fily, M. (2004). Climate variability along latitudinal and longitudinal transects in East Antarctica. Annals of Glaciology, 39, 351–358. https://doi.org/10.3189/172756404781813961
- Martin, P. J., & Peel, D. A. (1978). The spatial distribution of 10 m temperatures in the Antarctic peninsula. *Journal of Glaciology*, 20(83), 311–317. https://doi.org/10.3189/S0022143000013861
- McKay, M. D., Beckman, R. J., & Conover, W. J. (1979). A comparison of three methods for selecting values of input variables in the analysis of output from a computer code. *Technometrics*, 21(2), 239. https://doi.org/10.2307/1268522
- Morlighem, M., Rignot, E., Binder, T., Blankenship, D., Drews, R., Eagles, G., et al. (2019). Deep glacial troughs and stabilizing ridges unveiled beneath the margins of the Antarctic ice sheet. *Nature Geoscience*, 13(2), 132–137. https://doi.org/10.1038/s41561-019-0510-8
- Nias, I. J., Nowicki, S., Felikson, D., & Loomis, B. (2023). Modeling the Greenland ice sheet's committed contribution to sea level during the 21st century. Journal of Geophysical Research: Earth Surface, 128(2), e2022JF006914. https://doi.org/10.1029/2022JF006914
- Osher, S., & Fedkiw, R. P. (2001). Level set methods: An overview and some recent results. *Journal of Computational Physics*, 169(2), 463–502. https://doi.org/10.1006/jcph.2000.6636
- Pollard, D., & DeConto, R. M. (2012b). Description of a hybrid ice sheet-shelf model, and application to Antarctica. Geoscientific Model Development, 5, 1273–1295. https://doi.org/10.5194/gmd-5-1273-2012
- Pollard, D., & DeConto, R. M. (2020). Improvements in one-dimensional grounding-line parameterizations in an ice-sheet model with lateral variations (psuice3d v2.1). *Geoscientific Model Development*, *13*(12), 6481–6500. https://doi.org/10.5194/gmd-13-6481-2020
- Reijmer, C. H., van den Broeke, M. R., Fettweis, X., Ettema, J., & Stap, L. B. (2012). Refreezing on the Greenland ice sheet: A comparison of parameterizations. *The Cryosphere*, 6(4), 743–762. https://doi.org/10.5194/tc-6-743-2012
- Ritz, C., Edwards, T. L., Durand, G., Payne, A. J., Peyaud, V., & Hindmarsh, R. C. A. (2015). Potential sea level rise from Antarctic ice-sheet instability constrained by observations. *Nature*, 528(7580), 115–118. https://doi.org/10.1038/nature16147
- Schoof, C. (2007). Ice sheet grounding line dynamics: Steady states, stability, and hysteresis. Journal of Geophysical Research, 112(F3), 1–19. https://doi.org/10.1029/2006JF000664
- Seguinot, J. (2013). Spatial and seasonal effects of temperature variability in a positive degree-day glacier surface mass-balance model. *Journal of Glaciology*, 59(218), 1202–1204. https://doi.org/10.3189/2013JoG13J081
- Shapiro, N. M., & Ritzwoller, M. H. (2004). Inferring surface heat flux distributions guided by a global seismic model: Particular application to Antarctica. *Earth and Planetary Science Letters*, 223(1–2), 213–224. https://doi.org/10.1016/j.epsl.2004.04.011
- Sun, S., Cornford, S. L., Moore, J. C., Gladstone, R., & Zhao, L. (2017). Ice shelf fracture parameterization in an ice sheet model. *The Cryosphere*, 11(6), 2543–2554. https://doi.org/10.5194/tc-11-2543-2017
- Tsai, C.-Y., Forest, C., & Pollard, D. (2020). The role of internal climate variability in projecting Antarctica's contribution to future sea-level rise. *Climate Dynamics*, 55(7–8), 1875–1892. https://doi.org/10.1007/s00382-020-05354-8
- Whitehouse, P. L., Wiens, D. A., Gomez, N., & King, M. A. (2019). Solid earth change and the evolution of the Antarctic ice sheet. Nature Communications, 10, 1–14. https://doi.org/10.1038/s41467-018-08068-y
- Winkelmann, R., Martin, M. A., Haseloff, M., Albrecht, T., Bueler, E., Khroulev, C., & Levermann, A. (2011). The Potsdam parallel ice sheet model (pism-pik) – Part 1: Model description. *The Cryosphere*, 5(3), 715–726. https://doi.org/10.5194/tc-5-715-2011